Author's response to reviews

Title: A simple reproducible method of preventing lobar torsion

Authors:

MANOJ M PUROHIT (drpurohitm@yahoo.com)
JOSEPH J ZACHARIAS (Mr.Zacharias@bfwhospitals.nhs.uk)

Version: 2 Date: 25 April 2008

Author's response to reviews:

To,
The Editor-in–Chief,
Journal Cardiothoracic Surgery

Sub: Cover Letter for revised submission

Dear Editor,

We are re-submitting this manuscript ‘A Simple Reproducible Method Of Preventing Lobar Torsion’ for publication in your esteemed journal.

I have done following changes as asked.
Title-I have changes the case of title.
Author list - I have removed the qualifications, we want same sequence MP first, JZ 2nd.
I have removed the word count from the title page and key words from the abstract.
I have included a background header above the start of the main text on page 3.
I have included authors’ contributions - towards the end of the manuscript text, directly preceding the acknowledgements.
There are no acknowledgements, funding or conflicts of interest for this article.
I have done the correction in references.

Thanks
Yours truly,
Manoj Purohit
Mr. Manoj Purohit
7, Hilary Drive
Upton